Author's response to reviews

Title: Factors influencing Professionalism in nursing among nurses in Mekelle Public hospitals, Northern Ethiopia, 2012

Authors:

Atsede F Aregay (atsedefantahun@gmail.com)
Asrat D Megenta (asrat.megemta@yahoo.com)
Kahsu G Gebrekidan (kahsit12@yahoo.com)
Ayalnesh Z Yalew (ayuzem21@yahoo.com)
Gebre Y Abyu (gebyit45@yahoo.com)

Version: 3 Date: 20 August 2013

Author's response to reviews: see over
Factors influencing Professionalism in nursing among nurses in Mekelle Public hospitals, Northern Ethiopia, 2012

Atsede Fantahun*1, Asrat Demessie2, Kahsu Gebrekirstos3 Aynalnesh Zemene4, Gebre Yetayeh5,

1. Mekelle University, College of Health Sciences, Department of Nursing, e-mail: atse2000m@yahoo.com
2. Addis Ababa University, College of Health Sciences, Department of Nursing and Midwifery e-mail: asrat.megenta@yahoo.com
3. Mekelle University, College of Health Sciences, Department of Nursing e-mail: kahsit12@yahoo.com
4. Mekelle University, College of Health Sciences, Department of Midwifery e-mail: ayuzem21@yahoo.com
5. Mekelle University, College of Health Sciences, Department of Nursing e-mail: gebyit45@yahoomain.com

*Is the Principal investigator of this article.

I need to publish this thesis to disseminate the results to the public and to provide awareness about the factors which influence professionalism in nursing to all the society.